Author's response to reviews

Title: Quantitative measurements of the inequality in geographic accessibility to pediatric care in Oita Prefecture, Japan: Standardization with complete spatial randomness

Authors:

Susumu Tanimura (aruminat@gmail.com)
Masayuki Shima (shima-m@hyo-med.ac.jp)

Version: 2 Date: 2 May 2010

Author's response to reviews:

We are sending the revised manuscript entitled “Quantitative measurements of the inequality in geographic accessibility to pediatric care in Oita Prefecture, Japan: Standardization with complete spatial randomness”, which we would like to submit for publication as an original research article in the BMC Health Services Research.

In this study, we proposed a new inequality measure in geographic accessibility to pediatric care, adjusting the background geographic distribution pattern.

We sincerely thank for your fruitful comments.

The manuscript has not been previously published, nor is it under consideration for publication elsewhere.

I would very much appreciate for your kind consideration for the acceptance in publication.

Sincerely yours,